Author's response to reviews

Title: Reduced Crying in Term Infants Fed High Beta-Palmitate Formula: A Double-Blind, Randomized, Clinical Trial

Authors:

Ita Litmanovitz (litmani@clalit.org.il)
Fabiana Bar-Yoseph (fabiana@ENZYMOTEC.COM)
Yael Lifshitz (Yaell@ENZYMOTEC.COM)
Keren Davidson (davidson.keren@gmail.com)
Alone Eliakim (Eliakim.Alon@clalit.org.il)
Rivka regev (regevr@clalit.org.il)
Dan Nemet (Dan.Nemet@clalit.org.il)

Version: 2 Date: 1 January 2014

Author's response to reviews:

31-Dec-13

Dear Ms. Pura
Journal Editorial Office
BioMed Central

Thanks for your help with my manuscript submission.
I made the requested changes:
1. The title page – I added the email address of all authors and moved the authors' details at the bottom of page 15.
2. I added (after the conclusion)
   a. Abbreviations - page 13
   b. Conflict of interest - page 14
   c. Authors contributions - page 14
   d. Acknowledgement - page 14
I Hope this paper in its current form meets your requests

Sincerely yours
Ita Litmanovitz MD